Author’s response to reviews

Title: Cross-cultural adaptation of the Pain Medication Questionnaire for use in Brazil

Authors:

Sheila Galindo (sheilagalindo@hotmail.com)
Manoel Henrique Marinho (marinho75@poli.br)
Robert Gatchel (gatchel@uta.edu)
Everton Sougey (evertonbs@yahoo.com)
Eduardo Henrique Viana (eduardo.soares199@gmail.com)
Selene Vasconcelos (selumares@gmail.com)
Murilo Lima (murilodclima@gmail.com)

Version: 4 Date: 07 May 2019

Author’s response to reviews:

There is no text overlay, just a comparison with the original table and the one adapted for use in Brazil.